Author’s response to reviews

Title: Increased expression of neuregulin 1 in the urothelium of rat bladder with partial bladder outlet obstruction

Authors:

Seung Woo Yang (white0941@hanmail.net)
Seong Woo Jeong (swjeong@yonsei.ac.kr)
Ki Hak Song (urosong@cnu.ac.kr)

Version: 2 Date: 13 Nov 2017

Author’s response to reviews:

Dear reviewers,

Thank you for reviewing our manuscripts.

1. Euthanasia

Please clarify in the methods section of your manuscript if the rats were euthanized after the study finished. please also state the method of euthanasia.

-> Method section, line 16, page 4

2. Authors’ Contributions

Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. For example, the initials of a person named John Smith would be ‘JS’. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors, for example, by adding their middle name initial.

-> Author contributions section, page 9

3. Availability of data and materials

As part of our editorial policies we strongly encourage all authors to deposit their sequencing data before publication. Please deposit the sequencing data generated in this study in a suitable
public repository such as the NCBI SRA database. Once you have deposited your data, ensure you provide the information on deposition and how to access these data in the ‘Availability of Data and Materials’ section in Declarations, including the permanent link or the unique identifier associated to it.

-> The datasets used and analysed during the current study are available from the corresponding author on reasonable request.

4. Figures

Please remove the figure legends embedded within the figure files. All figure titles/legends should be listed and placed at the end of the main manuscript, after the References, and not within any of the figure files.

-> page 13

5. Declarations heading

Please add the heading "declarations" before the Declarations subheadings listed in your manuscript.

-> page 9

6. Clean copy

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

------------------------------------------------

Manuscripts has been carried out at the correction site recommended by BMC urology, and the proof is attached.

Thank you.

Sincerely yours

2017. 11. 13